

Article



Indicators for Ex-Post Evaluation of Cultural Heritage Adaptive Reuse Impacts in the Perspective of the Circular Economy

Martina Bosone ¹, Pasquale De Toro ², Luigi Fusco Girard ^{1,2}, Antonia Gravagnuolo ¹ and Silvia Iodice ^{1,3,*}

- ¹ Institute for Research on Innovation and Services for Development, National Research Council (CNR-IRISS), 80134 Naples, Italy; m.bosone@iriss.cnr.it (M.B.); l.fuscogirard@iriss.cnr.it or girard@unina.it (L.F.G.); a.gravagnuolo@iriss.cnr.it (A.G.)
- ² Department of Architecture, University of Naples Federico II, 80134 Naples, Italy; pasquale.detoro@unina.it
- ³ Joint Research Centre, European Commission, 21027 Ispra, Italy
- Correspondence: s.iodice@iriss.cnr.it

Abstract: Cultural heritage (CH) is considered a key element of cities and regions' identity and uniqueness, contributing to peoples' wellbeing and health, as well as jobs creation, environmental regeneration and place attractiveness. The adaptive reuse of abandoned and underused CH can be a sustainable strategy for heritage conservation, stimulating local development processes. However, heritage conservation needs large investments, while the resources available are scarce, and investment projects are subject to high uncertainties. Therefore, a careful assessment of impacts is needed to orient and direct CH adaptive reuse projects towards sustainability. Recent studies approach the adaptive reuse of abandoned buildings and sites as an effective circular economy strategy, potentially contributing to climate objectives through environmental regeneration and the reduction of natural resources consumption. However, evaluation tools to assess the impacts and orient adaptive reused interventions in the perspective of circularity are lacking. Through the analysis of 76 literature sources on CH impacts, this article explores how indicators are currently used in CH research and practice as impact assessment tools. More than 3500 indicators were retrieved and classified. Finally, this article proposes a comprehensive evaluation framework to assess the impacts of cultural heritage adaptive reuse in the perspective of the circular economy. The results showed that, while some indicators are available, many circularity aspects are not considered in the current studies on CH impacts.

Keywords: multidimensional indicators; evaluation tools; cultural heritage; adaptive reuse; circular economy; circular city; impact assessment; built environment; sustainable development

1. Introduction

Cultural heritage is considered a key element of city and region identities and uniqueness, potentially contributing to peoples' wellbeing and health, as well as to job creation, environmental regeneration and place attractiveness [1–5]. However, heritage conservation needs large investments, while the resources available are scarce and investment projects are subject to high uncertainties [6–9].

The adaptive reuse of abandoned and underused cultural heritage sites can be a strategy to enhance heritage conservation, stimulating sustainable development processes through new uses of old buildings and sites, co-creating new meanings and re-activating neglected areas, turning them into new vibrant cultural places [10–13]. Adaptive reuse is defined as "any building work and intervention to change its capacity, function or performance to adjust, reuse or upgrade a building to suit new conditions or requirements" [14]. Several authors [10,15,16] have stressed the importance of adaptive reuse for urban regeneration. Günçe and Mısırlısoy [17] explored how investments in adaptive reuse can contribute to revitalise neglected areas, thus improving the living standards for the local community and attracting consequently new investments that foster economic growth in a virtuous circle, enhancing sustainable urban development [18].



Citation: Bosone, M.; De Toro, P.; Fusco Girard, L.; Gravagnuolo, A.; Iodice, S. Indicators for Ex-Post Evaluation of Cultural Heritage Adaptive Reuse Impacts in the Perspective of the Circular Economy. *Sustainability* 2021, *13*, 4759. https://doi.org/10.3390/su13094759

Academic Editor: Asterios Bakolas

Received: 9 February 2021 Accepted: 9 April 2021 Published: 23 April 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). According to the recent literature, the adaptive reuse of abandoned and underused buildings and sites, which represent urban "wastes", also supports the implementation of the circular economy model in the spatial dimension [12,19,20].

The interrelations and boundaries between the two concepts of sustainability and circular economy have been addressed in the scientific literature [21], while a clear and agreed definition of circular economy has not yet been established [22,23]. According to the review of literature developed by Geissdoerfer et al. [21], circular economy can be defined as "a regenerative system in which resource input and waste, emission, and energy leakage are minimised by slowing, closing, and narrowing material and energy loops. This can be achieved through long-lasting design, maintenance, repair, reuse, remanufacturing, refurbishing, and recycling." Additionally, Sustainability can be defined as "the balanced integration of economic performance, social inclusiveness, and environmental resilience, to the benefit of current and future generations". In this paper, circular economy is interpreted as a new economic model, the implementation of which is instrumental to achieve long-lasting sustainability.

The current reviews of the literature on circular economy indicators showed that the most used indicators are related to waste management, raw materials, recycling rates, economic performance of circular businesses, energy, toxicity and clean material cycles [24,25]. A different study focused on the objectives of using circular economy indicators, classifying them based on a taxonomy of ten categories: levels, loops, performance, perspective, usage, transversality, dimension, units, formats and sources [26]. In particular, the social and cultural dimension of circular economy seem to be less explored in the studies analysed, providing less consideration on the impacts of circular economy initiatives on people and local communities [22,27,28]. Some initial studies indicate that implementing circular economy [29] may generate positive impacts on human health, pointing out the existence of a relationship between human health and climate change issues [30–33].

Even though circular economy started to be implemented in the built environment and building construction sectors in recent years [34,35], circular economy indicators are rarely applied to the specific cultural heritage sector [36,37]. The concept of cultural heritage adaptive reuse as an instrument to achieve circular economy goals at the territorial scale emerged more recently, but indicators linked to heritage conservation are quite often only linked to the environmental dimension, while integrated and multidimensional approaches are still lacking [11,12].

All in all, when it comes to the relationships between cultural heritage adaptive reuse and circular economy in general, few studies are currently available. Indirect approaches can be identified in the ecosystems services assessment frameworks [38-40]. For example, Stanik et al. [41] analysed cultural heritage from the perspective of cultural ecosystem services, with the aim of identifying and developing an indicator-based framework formed by indicators related to historic land uses and historic elements. Still, Gravagnuolo et al. proposed an evaluation framework for circular economy implementation in the adaptive reuse of cultural heritage [12], while Foster [42] associated the concepts of adaptive reuse, cultural heritage and circular economy, focusing on the environmental benefits of heritage conservation. More in depth, this paper performs a literature review to demonstrate the alignment between circular economy goals and adaptive reuse of heritage assets in a life-cycle perspective. Foster and Kreinin [43] also performed an in-depth review of environmental indicators for the adaptive reuse of cultural heritage in the perspective of the circular economy, able to demonstrate the environmental savings of adaptive reuse. Then, Foster et al. [44] developed a Circular Environmental Impact Indicator Framework for cultural heritage adaptive reuse in order to integrate macro-European Union-level indicators with environmental indicators at the micro scale.

Despite the huge range of studies developed in order to assess the multidimensional impacts of cultural heritage conservation, most of these studies focus on single impacts, for example cultural tourism, cultural and creative sectors, sustainability, wellbeing, while an integrated perspective is still missing and considerable efforts are still necessary [12].

Moreover, the decision-making processes for cultural heritage requires careful attention and cannot ignore the use of appropriate decision-making tools, hence in this context "evaluation can be considered a relevant tool to build choices, to recognise values, interests and needs, and to explore the different aspects that can influence decisions" [45]. Consequently, evaluation methods facilitate the decision-making process when different solutions are available, but different criteria have to be taken into account and the involved decision-makers may be strongly conflicting [46].

Moreover, the adaptive reuse of abandoned buildings and sites can represent a substantial contribution to circular economy implementation in cities and regions, if adequate design and reuse choices are taken, promoting environmental regeneration and avoiding over-consumption of soil, raw materials and other natural resources [12,42].

As there is a high potential for synergic outcomes, investments in cultural heritage adaptive reuse can be effectively directed towards circularity to actually generate net positive economic, social, environmental and cultural impacts in the territories [12,47]. However, evaluation tools to assess the impacts and orient adaptive reuse interventions in the perspective of the circular economy implementation in cities and regions are lacking [48,49].

1.1. Evaluation for Cultural Heritage Adaptive Reuse and the Need for New Criteria and Indicators

The concept of evaluation for cultural heritage and the general development of thematic criteria and indicators emerged since the late 1990s on an international level [50], and this is still confirmed by the more or less recent development of many documents and international papers [15,23–29].

However, despite the large body of literature on heritage evaluation tools and indicators, there is still confusion between terms such as "criteria" and "indicators". In this paper, the definition proposed by Milan Zeleny and recalled in Gravagnuolo et al. [12] is adopted, identifying "criteria" as the specific "points of view" through which a goal or objective is analysed and "indicators" as the elements through which criteria are assessed or "measured"—meaning, with "measure", both qualitative and quantitative assessments.

The objectives of using multidimensional criteria and specific indicators are diverse, from mapping and assessment of heritage attributes and values to ex-ante evaluations to assess heritage conservation vs. transformation choices to ex-post evaluations focusing on the actual impacts generated through heritage investments. In particular, ex-post evaluation is used in order to verify that the established objectives have been met, determining if there are unforeseen or unintended consequences and assessing in the meantime the performance of alternative approaches [51].

Della Spina [16] for example uses multi-criteria techniques in order to evaluate different adaptive reuse strategies, selecting a suitable set of indicators, pointing out the consideration of cultural heritage as a driver of urban development, as already also underlined by Guzmán et al. [52]. Elsorady [53] identifies a set of indicators with the goal of evaluating the compatibility of new uses for the adaptive reuse of heritage buildings, while Kutut et al. [54] analyse indicators to assess whether or not historic buildings should be reconstructed, where reconstruction is meant both as restoration of primary volumes but also preservation of as many authentic elements as possible. Stanik et al. [41] investigate the relationship between cultural heritage and cultural ecosystem services, developing an indicator-based framework aimed at mapping cultural heritage in the spatial dimension. Moreover, Nijkamp [55] gives an overview of cultural heritage evaluation methods, while Nocca [48] deepens the role of cultural heritage for sustainable development, stating that there is an insufficient amount of indicators that demonstrate this relationship.

The European research "Cultural Heritage Counts for Europe" analysed a large literature body on methods and tools for the assessment of impacts of cultural heritage conservation projects, highlighting diverse areas of impact based on the four pillars of sustainability: economic, social, environmental, cultural [56]. The reports of the "Heritage Counts" initiative developed in UK similarly explored the economic impacts of cultural heritage, deepening also wellbeing aspects [57,58]. Diverse studies focused on indicators to place cultural heritage in the sustainable development agenda [48,49,59,60]. More recently, a research carried out by the "European Spatial Planning Observation Network" (ESPON) underlined the role of Material Cultural Heritage (MCH) as a strategic driver for sustainable territorial development, where MCH is meant as "Objects including different types of immovable (e.g., archaeological sites, cultural landscapes, etc.) and movable (e.g., paintings, books, etc.) MCH recognised as having heritage value in each country [...]" [61]. The ESPON research proposes three key indicators to assess the economic impacts of cultural heritage conservation, valorisation and reuse activities: Gross Value Added (GVA) of the share related to MCH; Turnover of the share related to MCH; Number of employees of the share related to MCH. In addition, the study also considered the following indicators to complement the analysis: Value of heritage volunteering; and Expenditure by the public sector on MCH (investments by public authorities on cultural services and spending on conservation, repair and maintenance of protected constructions).

1.2. Aim and Structure of the Paper

The existing literature body on the linkages between cultural heritage adaptive reuse and circular economy shows a potential in building a comprehensive framework of quantitative and qualitative indicators to assess performances and impacts of adaptive reuse interventions in a circular economy perspective.

Despite many indicators are already in use in the cultural heritage sector and attempts to systematise heritage indicators have been widely developed [48,49,59], the circular economy perspective still needs to be fully implemented. This paper aims to fill this gap by analysing and classifying existing heritage indicators, comparing them with circularity criteria. Through the analysis of 76 literature sources on cultural heritage impacts, this article explores how indicators are currently used in heritage research and practice as impacts assessment tools. More than 3500 indicators were retrieved and classified. Finally, this article explores the concept of circular economy in cultural heritage adaptive reuse, identifying specific impact criteria based on recent scientific literature, highlighting the knowledge gaps for future research.

The following Section 2 describes the methodology developed to analyse and systematise existing heritage indicators, while Section 3 describes the results of the analysis and proposes specific criteria for the future development of heritage indicators in the perspective of the circular economy. Section 4 discusses the results and the knowledge gaps emerging from this study. Section 5 provides critical conclusions, focusing on the limitations of this study and the potential for future research.

2. Methodology

2.1. What Is an Indicator? Some Basic Definitions

The methodology applied was based on previous studies focusing on indicators review, especially Saidani et al. [26], adapted to the specific aim of this research. First, the diverse definitions of indicator were explored, to prepare the successive phase of literature search, analysis and systematisation.

The definition of indicator is not univocally recognised and the nuances between "indicators", "criteria", "objectives" and "attributes" may be not always clear in the applications. Indicators can be interpreted at different decision levels and are always linked to a set of "criteria", meant as "points of view" recognised as relevant [12]. Finally, indicators can be used to assess impacts, i.e., any change from an initial condition to a subsequent condition and they are referred to ex-post evaluation. They can also be used to analyse the state of a certain aspect (ongoing evaluation) and finally they can also support the decision-making phase before the implementation of a certain project (ex-ante evaluation).

Heink and Kowarik [62] propose the following definition of indicator as a "measure or component from which conclusion on the phenomenon of interest (the indicandum) can be inferred". They underline the necessity of avoiding failure in defining technical terms, proposing a systemic overview of existing definitions of the term indicator, with a special focus on ecology and environmental planning. According to Hockings et al. [63], indicators are "measurable entities that are used to assess the status and trend of a range of site values. A given value, objective, or additional information need can have multiple indicators. A good indicator meets the criteria of being measurable, precise, consistent and sensitive". Indicators represent also essential tools to synthesise complex information on the territorial functions, as they are able to monitor the state and the functioning of urban areas and to verify the achievement of predetermined goals [64]. Still, Zhang et al. [65] state that "indicators and measurement systems are an essential tool for ensuring management targets are reached [...]" and this was also confirmed by Stanners and Bordeau [66], who consider indicators as vital elements for developing awareness on urban issues.

Elaborating good indicators facilitates the decision-making phase, identifying limits and opportunities and thus reducing risks or costs. Through indicators, emerging issues and impacts can be envisaged, allowing corrective actions when necessary. Experts involved in the indicators elaboration are entrusted with the task of providing information that is comprehensible and credible by all for their correct use in the decisionmaking process [67].

Despite a huge quantity of definitions is available, in the present study we adopted the description of an indicator as «a quantitative or qualitative factor or variable that provides a simple and reliable means to measure achievement, to reflect changes connected to an intervention, or to help assess the performance of a development actor» [68]. Indicators must always be clearly defined in theoretical and operational terms and they must be simple and understandable according to their scope.

2.2. Analysing and Systematising Heritage Indicators

After defining what is an indicator and for which scopes it may be used, the relevant literature on heritage indicators was collected and analysed. The methodology applied was based on two main phases, articulated into diverse steps:

- Phase 1: existing studies on indicators for cultural heritage impacts assessment were collected, analysed and classified following four main steps, to assess how indicators are currently used in research and practice.
- Phase 2: the conceptual framework of "Circular CHAR (Cultural Heritage Adaptive Reuse)" was built based on the analysis of previous studies on circular economy models in cultural heritage conservation and adaptive reuse; previous studies were collected, analysed and synthesised to propose a comprehensive framework of circularity criteria for CHAR impacts assessment in the perspective of the circular economy and circular city model; finally, circularity criteria were compared with existing heritage indicators sources, identifying knowledge gaps for further research.

Figure 1 summarises the methodological process highlighting the main phases and steps.



Figure 1. Methodological process to select, synthesise, analyse and systematise heritage indicators and build the Circular Cultural Heritage Adaptive Reuse (CHAR) impacts assessment framework.

2.2.1. Phase 1: Analysis and Classification of Existing Heritage Indicators

"Heritage indicators" are here meant as indicators directly or indirectly related to cultural heritage impacts. They were selected from existing studies, including scientific articles retrieved from Scopus/WoS, Google scholar and Italian scientific journals indexed by the national agency for research quality evaluation (ANVUR). Additionally, "grey" literature was collected from institutional sources, including both policy-related documents and practice-based guidelines. Some examples are documents by ICOMOS, UNESCO, European Commission, Historic England, as well as international organisations focused on circular economy such as Ellen MacArthur Foundation, ARUP, Club of Rome, and others. The selection process consisted of using specific keywords within the "Science Direct" database (up to the year 2020) to facilitate the search, such as: "cultural heritage adaptive reuse" (1.621 initial results), "cultural heritage sustainability indicators" (9.754 initial results) and "cultural heritage indicators" (16.170 initial results). To narrow the focus of our search, literature sources were selected giving priority to the most recent (years 2017-2020). Sources that did not present indicators and were not strictly related to impacts assessment were immediately excluded. An in-depth analysis of the identified literature sources was then performed, reading the abstracts and selecting the most relevant documents for further analysis. Subsequently, titles, abstracts and keywords were carefully read in order to select the relevant literature to be analysed in-depth. Starting from this initial sample of about 20 documents, the references of these sources were analysed to identify further documents that could be useful for our purpose and adding also grey reports retrieved from the web.

Initial reading was processed by selecting papers and studies that focused on the evaluation of impacts of cultural heritage conservation, presenting a set of evaluation tools, methods, criteria and/or indicators. As existing reviews of indicators were included, this study has the characteristics of a systematic review integrated with a meta-analysis of diverse sources, where the meta-analysis specifically refers to "the statistical analysis of the data from independent primary studies focused on the same question, which aims to generate a quantitative estimate of the studied phenomenon" [69].

The methodological process in Phase 1 is described below, with reference to Figure 1 and to the three steps indicated.

Step 1: Selection and analysis of studies specifically focused on cultural heritage impacts, including existing reviews of indicators. This first screening allowed to select 23 papers and studies focused on cultural heritage impacts. A comprehensive overview of the literature sources analysed in this step is presented in Appendix A. In analogy with the results of previous studies [25,26], a database of literature sources and indicators classifications was built.

Step 2: Selection and analysis of studies related to urban development and urban regeneration which include heritage indicators. Using the same procedure described in step 1, literature sources indirectly linked to cultural heritage or to some of its most relevant domains (for example cultural tourism or the creative and cultural sectors) were collected and analysed. These domains have been identified on the basis of the first screening of analysed sources. Only a sub-set of indicators linked to cultural heritage was retrieved and classified from the studies included in Step 2. These indicators can be adapted to assess relevant impacts related to cultural heritage adaptive reuse, such as cultural vibrancy in a certain area where a reused cultural building is located. A number of 48 literature sources was collected in Step 2 (see Appendix B).

Step 3: Selection and analysis of specific studies focusing on circular economy and circular city that include heritage aspects and indicators. Circular economy literature was analysed, particularly related to circularity in the built environment and building construction sector, circular city model, and the few specific studies available on "circular adaptive reuse of cultural heritage". According to this analysis, additional 5 sources and 54 indicators were retrieved (see Appendix C).

All the selected documents were thus classified based on the year of publication, type of source, and sector of implementation to provide a general overview of the studies regarding cultural heritage indicators and the indicators related to its impact sectors.

Step 4: Construction of a taxonomy of indicators. Existing heritage indicators were classified based on the taxonomy categories. A first tentative taxonomy of indicators was structured according to the objectives of the study and based on other taxonomies found in previous studies. In particular, Saidani et al. [26] proposed, between other categories, to classify indicators based on the "level" or "scale", "usage", "sustainability dimension", "transversality of the implementation sectors" and "type of sources". Additionally, Gravagnuolo et al. [12] proposed to identify relevant indicators of "circular adaptive reuse of cultural heritage", according to the "sustainability dimension", "typology" (quantitative or qualitative) and "scale of implementation". According to these reference studies and taking into account the specific objectives of this analysis, the final taxonomy was built.

The existing indicators included in these selected studies were analysed and classified according to the following categories (see Appendices A–C):

- Geographical scale (level): "macro" scale was assumed for indicators addressed at the national scale (NUTS 0) or regional scale (NUTS 2), "meso" scale for indicators at city level and "micro" scale for indicators at the scale of the heritage building or site.
- Evaluation phase (usage): "ex-ante evaluation indicators", when the assessment is performed before a specific project or reuse design is realised, in order to take informed choices between diverse alternatives, "monitoring indicators" when the assessment is performed during the implementation of a specific project and "ex-post evaluation indicators" used to examine the results of actions or activities and to compare the programmed measures with the actual results.
- Typology: qualitative, if based on perceptions or subjective and unquantifiable aspects (soft); and quantitative, if based on precisely measurable aspects (hard).
- Sustainability dimensions addressed: economic; social; cultural; and environmental according to the "four pillars" approach proposed by "Cultural Heritage Counts for Europe" research [56].

The classification of existing indicators was performed based on the classification of the original studies. For example, some studies proposed a list of indicators in each sustainability dimension, which was useful to classify the specific indicators within the proposed taxonomy. Additionally, some studies focused on the urban dimension, therefore the related indicators were classified in the "meso" scale according to the proposed taxonomy. Where the classification in original studies was uncertain, indicators were compared with similar ones included in other studies expressing a clear classification with regard to the proposed taxonomy, which helped to identify a thorough and robust classification. Following this process, each indicator selected was classified according to all taxonomy categories.

2.2.2. Phase 2: Circular CHAR Impacts Assessment Framework Based on Criteria and Indicators

Phase 2 is aimed at building a conceptual and operational framework to assess the impacts of cultural heritage adaptive reuse in the circular economy perspective. This phase was focused on the identification of circularity dimensions based on previous literature (theoretical level), as well as evaluation criteria (operational level). The potential for implementation of the resulting "Circular CHAR" impacts assessment framework was finally assessed comparing the proposed criteria with the existing heritage indicators analysed in Phase 1, in order to identify robust domains of implementation and knowledge gaps for further research.

The methodological process in Phase 2 is described below, with reference to Figure 1.

Step 5: in-depth reading of the selected studies on circular economy/circular city/circular buildings was performed, including both a thematic-analysis and a contentanalysis. This process led to the identification of the "Circularity dimensions", representing the basis of the impact assessment framework for "Circular Cultural Heritage Adaptive Reuse" (Circular CHAR). The "Tripod model" (see Section 3.2) developed within the Horizon 2020 research project "CLIC: Circular models Leveraging Investments in Cultural heritage adaptive reuse" [11,70,71] was the main reference for the "Circular CHAR" framework developed in this study at the theoretical level.

Step 6: based on the mentioned "Tripod model" adopted, as well as previous literature on circular adaptive reuse of cultural heritage, a structured set of circularity criteria was identified, resulting in the "Circular CHAR" impacts assessment framework. The evaluation criteria proposed in the literature were synthesised and discussed during a series of 5 Focus group sessions conducted with an interdisciplinary group of 14 experts in heritage conservation, heritage economics, evaluation methods, circular economy, sustainable finance, economic spillovers of development projects and social impacts, including CLIC project researchers. Focus groups were conducted through in-person and online meetings between January and July 2020. During each focus group, an initial long list of proposed evaluation categories and related indicators retrieved from literature sources was discussed. Each expert was provided with preparatory documents and was asked to provide comments in written form for the next experts' meeting. During the five focus group sessions, the list of criteria was refined and synthesised to avoid overlapping and specify descriptions and definitions.

Step 7: the circularity criteria identified in Step 6 were compared with the existing studies and available heritage indicators analysed and classified in Phase 1, to assess the level of potential operationalisation of the Circular CHAR framework.

The following section describes in detail the results of each phase of the study.

3. Results

3.1. Results of Phase 1: Building the Heritage Indicators Literature Database

The analysed literature sources were classified according to the following criteria:

- Year of publication.
- Typology: institutional reports (i.e., report provided by Institutions), scientific articles, research reports, working papers, websites, handbooks and books.
- Scope: ranging from the sources strictly focused on cultural heritage impacts (cultural heritage), to sources addressing other sectors indirectly associated to cultural heritage impacts ("other topics") and, finally, to sources linked to the concept of circular economy applied to cultural and built heritage.



Figure 2. Temporal frequency of the analysed sources.

A total of 76 literature sources was analysed. Figure 2 shows the temporal frequency of publications from the 80s to 2020. It can be recognised an increasing attention to heritage indicators over time, with the highest frequency in the last ten years. This attention is growing, with a peak in the more recent years (2014–2020).

The typology of literature sources analysed showed a significant prevalence of institutional reports (about 35.5%) and scientific articles (32.9%). Research technical reports represented about the 22.3% of the total sources collected, while working papers, books and websites were present only for a small percentage (between 2.6 and 1.3%). In particular, websites were used as a source from which analysed indicators were directly deduced. These are online information and monitoring platforms used by international associations and organisations to assess and support cultural heritage strategies.

The "Compendium of Cultural Policies & Trends" [72] is an online database with in-depth information on cultural policies, statistics and trends. It shows two monitoring categories, "Statistics" and "Comparison", each of which is subdivided into sub-categories from which the indicators have been derived directly.

The Eurostat "Circular Economy Indicators" [73] is a monitoring framework set up by the European Commission. The framework consists of ten indicators, some of which are broken down in sub-indicators, for a total of 16 indicators. Compared to the four categories suggested in the framework ("Production and Consumption", "Waste Management", "Secondary Raw Materials" and "Competitiveness and Innovation") the indicators were reorganised and adapted to the four dimensions proposed in this research.

Academic and practice literature was included, so that the database is representative of theoretical and practical knowledge (Figure 3a).



Figure 3. Classification of the analysed sources according to (a) Typology; (b) Scope.

Literature sources were classified according to their scope. The first group was made of 23 sources specifically related to "Cultural heritage" (30%). The second groups included 48 sources not specifically related to cultural heritage, but in which heritage-related indicators are present, classified as "Other topics" (63%). As already specified, for those sources, only a sub-set of indicators was analysed. Indeed, indicators were selected both from sources strictly linked to cultural heritage, and from sources linked to broader sectors, such as circular built environment. In this last case, studies and indicators useful to evaluate the circularity of cultural heritage adaptive reuse have been selected, according to the boundaries of the investigation as set out in the methodology section. Finally, a number of 5 sources related to "Circular economy & cultural heritage", as well as built environment was identified (7%), in which all indicators were considered for the successive analysis (Figure 3b).

A number of 3543 relevant indicators was collected and classified according to the proposed taxonomy: geographical scale, evaluation phase, typology and sustainability dimension addressed.

Figure 4 shows how the indicators are distributed and classified. In particular, Figure 4a shows that most of the indicators are applied at the macro scale of the region or state (52.6%), followed by indicators applied at the meso scale of the city and urban areas (34.2%) and a smaller percentage applied at the micro scale of the building or site (13.2%). While the adaptive reuse interventions on cultural heritage are mostly realised in specific buildings or sites, there is evidence of a lack of indicators supporting the evaluation of impacts at the "micro" scale. Regarding the evaluation phase, it has been identified according to the scope of the evaluation in the original sources. Figure 4b shows that most indicators are applied in the ex-post evaluation phase (94.6%), highlighting a particular focus of existing policies and practices in the post-realisation phase, according to the literature sources selected. Clearly, it is important to assess the impacts of heritage conservation, reuse and valorisation projects after their realisation. The ex-post assessment should also represent a starting point for future practices, learning from the past to take better choices. However, the scarce attention in the ex-ante and ongoing/monitoring evaluation seems to highlight that evidence-based evaluations are poorly adopted in the planning and design stages. The classification of indicators was conducted based on clear information retrieved from the reference sources, as well as comparison with other similar indicators used in other studies. However, in some cases the classification was uncertain, due to the unclear definition of the fields of application of the reference documents analysed. Therefore, some indicators were exclusively classifiable as "ex-post", while in other cases the classification of indicators usage was flexible and could be applied in different phases of the evaluation. It should also be noted that, in order to build effective decision-support tools, evidence-based data should be collected on extensive scale and following structured, harmonised and agreed approaches, building datasets that can support as far as possible the estimation of key indicators in the ex-ante design and planning phase. Quantitative indicators are needed in this sense and represent the 66.3% of all indicators included in the database, as showed in Figure 4c, while 33.7% are qualitative indicators based on "soft" data mostly represented by citizens and stakeholders' perceptions, as well as spatial or visual representations. Observing the sustainability dimensions addressed, it is possible to note that indicators related to the social dimension represent the majority (33.4%), economic indicators are also well represented (28.7%), and indicators related to the cultural dimension represent the third group (25.1%). Indicators related to the environmental dimension of heritage conservation are limited (12.8%), which highlights that the environmental impact of heritage conservation has been quite disregarded in the heritage sector, not representing a particular focus for researchers and practice stakeholders. However, as stated by Fusco Girard et al. [11,12,42], a circular economy approach in heritage conservation could substantially contribute to achieve climate objectives and reduce the overall costs of conservation, turning it into an "investment".

Phase 1 of the analysis resulted in a better understanding of the actual use of indicators in the heritage field as well as in heritage-related additional sectors. The database of literature sources analysed in this study is available as Supplementary Materials. Between the literature sources scope groups, the circular economy was identified as an emergent issue for cultural heritage research, since many studies related to circular economy have been proposed in the most recent years. Therefore, a set of specific dimensions and criteria of circularity was identified in Phase 2 to build a comprehensive impacts assessment framework for Circular CHAR.



Figure 4. Classification of indicators according to (**a**) Geographical scale; (**b**) Evaluation phase; (**c**) Typology; and (**d**) Sustainability dimensions.

3.2. Results of Phase 2: Circular CHAR Structured Framework of Criteria and Indicators

According to the existing literature on cultural heritage adaptive reuse in the perspective of the circular economy and circular city model, a structured set of evaluation criteria was developed, aimed to reflect the circular economy perspective in CHAR.

In Step 5, the diverse general definitions of circular economy were explored [22], identifying a specific definition of circular economy for cultural heritage adaptive reuse based on recent literature. One of the key studies analysed was Gravagnuolo et al., which defines the circular economy in cultural heritage adaptive reuse as "a sustainable economy that enables a continuous positive development cycles that preserve and enhance the created values, in an indefinite time, of cultural and natural capital, optimises resource yields and minimises system risks by managing finite stocks and renewable flows" [12]. In the mentioned study, the main characteristics of the circular economy model for cultural heritage were identified, including diverse frameworks such as the 9 Rs approach (Reuse, Reduce, Reuse, Repair, Refurbish, Remanufacture, Repurpose, Recycle, Recover) [74,75], and the ReSOLVE framework proposed by Ellen MacArthur Foundation (Regenerate, Share, Optimize, Loop, Virtualize, Exchange) [76–78]. Therefore, according to Gravagnuolo et al. and other relevant scientific articles [12,79,80], a first set of evaluation criteria for CHAR was developed, highlighting potential and actual impacts of CHAR projects in the eco-

nomic, social, cultural and environmental dimension. The following dimensions of circular economy implementation in CHAR were initially identified:

- Conservation and regeneration of cultural heritage values—linked to cultural capital regeneration, both tangible and intangible.
- Circular economy implementation in building/site recovery works—linked to natural capital regeneration through technologies for energy, water, wastes, materials, including nature-based solutions.
- Impacts in the local context—including jobs creation, attractiveness of the place for businesses, cultural tourism, residents, commercial and artisanal activities.
- Circular business, financing and governance models applied—linked to inclusiveness in the decision and management process engaging local communities, capacity of generating revenue flows able to cover investments and management costs, sustainability of financing sources.

According to the research conducted under the Horizon 2020 CLIC project, the conceptual framework of the "Tripod model" was analysed. A further study of Gravagnuolo, Fusco Girard, Kourtit and Nijkamp [70] specified the conceptual evaluation framework placing CHAR in the perspective of the circular city model, identifying three main critical drivers or "building blocks" of circularity:

- a "regenerative capacity" [70] linked to the self-regeneration of the cultural assets, as well as of the economic, environmental and social resources needed for its maintenance over time (in analogy with the circular economy principle of extending the use value of resources in the largest time horizon possible).
- a "generative capacity" [70], linked to the net positive economic, environmental and social externalities generated in the area/territory—which in part come back to the heritage asset.
- a "symbiotic capacity" [70], linked to the cooperation and collaboration approaches that enable a more efficient use of resources (such as those realized in "industrial symbioses"), as well as clustering processes in the territory (implementing an "economy of relationships").

Thus, the work conducted under the CLIC research project was integrated in the conceptual framework presented in this article. According to Luigi Fusco Girard, the auto-poietic model of nature regeneration is embedded in the theoretical framework of circular adaptive reuse of cultural heritage [11,71,81]. As natural systems, the heritage site can be interpreted as a "lively regenerative/auto-poietic system", able to self-generate the resources needed for its functioning and to use all wastes as resources for new productive cycles [71]. Through the circular economy approach applied, it is possible to interpret and evaluate the adaptive reuse process identifying diverse forms of capital that are regenerated: man-made capital, natural capital, social capital, human capital. The model of circular CHAR was initially tested through empirical evidence based on a large dataset of 126 projects in CHAR, identifying a set of "building blocks" of circularity through statistical methods: Cultural value, Management characteristics, Circular metabolism, Landscape quality, Social impact, Economic spillovers [82].

Based on the conceptual evaluation model of "Circular CHAR", and its initial implementation through case studies analysis, a structured set of evaluation criteria for circular adaptive reuse of cultural heritage was built. The "circularity dimensions" as proposed by Luigi Fusco Girard as the "Tripod model" and recalled in Gravagnuolo, Fusco Girard, Kourtit and Nijkamp [70] were considered:

- The "regenerative capacity"
- The "symbiotic capacity" and
- The "generative capacity".

Once defined the main circularity dimensions based on the "Tripod model", in Step 6, an initial long list of evaluation criteria was developed. Circularity criteria were thus

discussed during a series of 5 Focus group sessions conducted with an interdisciplinary group of 14 experts including CLIC project researchers.

A number of 40 resulting criteria resulted from different rounds of discussion. A description was provided for each criterion (Table 1). Finally, the circularity criteria were compared with existing indicators retrieved and classified in the first phase of the analysis. The level of operationalisation of the impact assessment framework was assessed based on three aspects considered:

- Criteria for which well-established evaluation methods and indicators are available, based on the meta-analysis of heritage impacts studies (visualized with symbol "O" in Table 1).
- Criteria for which evaluation methods and indicators could be available from existing sources but should be adapted to the circularity framework (visualised with symbol "Δ" in Table 1).
- Criteria for which well-established evaluation methods and indicators are currently not available, or only few studies could be detected in recent years, showing a knowledge and implementation gap with regard to existing heritage indicators potentially applicable. These criteria may actually represent a new field of study for circularity and cultural heritage, which can be further developed/enhanced within the Circular CHAR: Intrinsic value, Local circular economies, Heritage community, Circular metabolism, Smart Specialisation Strategies, Wellbeing and Health impacts (visualised with symbol "X" in Table 1).

Circularity Dimension	Criteria	Sustainability Dimension	Description/Motivation	Comparison with Current Indicators
	Authenticity and integrity	Cultural	Regeneration of cultural capital, tangible and intangible, through conservation of heritage authenticity and integrity as defined by UNESCO and ICOMOS	Δ
Regenerative capacity (auto-poietic capacity)	Intrinsic value	Cultural	Re-generation and transmission of heritage values and meanings through the adaptive reuse intervention, also through hybridisation between historic and contemporary values integrated with cultural landscape and coherent with the intrinsic value of cultural heritage	х
	Financial self-sustainability	Economic	Self-generation of financial resources needed for heritage conservation and continuous maintenance, through diverse revenue flows from reuse activities; independence of financial sources from public sector	Δ
	Local circular economy	Economic	Circular re-use of profits in the local context for further adaptive reuse projects and/or activities in additional heritage buildings or sites, circular entrepreneurial activities, social and solidarity economy activities; Local resources such as food, craft, materials are valorised through the adaptive reuse; stimulation of local investments and economic activities linked to proximity	x
	Energy efficiency	Environmental	Self-generation of energy sources for the operational phase through renewables, also reducing energy consumption needs through heritage-compatible technologies	0
	Freshwater efficiency Environmenta		Self-generation of water resources for the operational phase through water capture, filtering and reuse systems	О
	Nature-Based Solutions	Environmental	Regeneration of natural resources through nature-based solutions aimed to enhance air quality, freshwater quality, green surfaces	0

Table 1. Evaluation framework of criteria for Circular Cultural Heritage Adaptive Reuse.

Circularity Dimension	Criteria	Sustainability Dimension	Description/Motivation	Comparison with Current Indicators
	Soil recovery	Environmental	Remediation of polluted soils and brownfields, land recovery through reuse interventions	О
	Heritage community	Social	Self-organisation capacity of active citizens to build a cohesive and pro-active Heritage Community for heritage conservation, valorisation and adaptive reuse	X
	Local community	Social	Enhancement of skills, education and learning opportunities for the local community	Δ
	Traditional skills	Cultural	Enhancement of traditional skills through the adaptive reuse, incl. rehabber approaches and training opportunities	Δ
	Local identity	Cultural	Contribution of the adaptive reuse intervention to local identity; enhanced access to the educational function of cultural heritage	Δ
	Mutual cooperation	Cultural	Attitude of stakeholders to mutual cooperation, sharing common resources, knowledge, assets; involvement of third sector actors and/or sharing economy actors in the adaptive reuse; Collaboration pacts/agreements between public, private and people are implemented	Δ
	Cultural and knowledge capital production	Cultural	Knowledge production and cultural production stimulated by the adaptive reuse intervention	Δ
Symbiotic capacity	Circular metabolism	Economic	Realisation of circular supply chains to reduce costs of energy, wastes, materials, water achieving circular metabolism of heritage buildings and sites	X
	Smart Specialisation Strategies	Economic	Contribution of the adaptive reuse intervention to regional development, through coherence of the reuse functions and processes with regional Smart Specialisation Strategies	Х
	Construction & Demolition Wastes	Environmental	Avoided Construction & Demolition Wastes through the adaptive reuse intervention	О
	Materials extraction	Environmental	Avoided raw materials extraction through the adaptive reuse intervention	О
	Participation in decision-making	Social	Involvement of diverse stakeholders and citizens in the decision-making process for the adaptive reuse	Δ
	Social cohesion	Social	Increase of trust and awareness of present and future generations' needs and rights	Δ
	Civic pride	Social	Enhancement of civic pride, belonging and civic responsibility; shared values and bonds in the local community, openness of the local community	Δ
	Cultural vibrancy	Cultural	Enhancement of cultural activities and events as a result of the adaptive reuse; enhanced access to culture and cultural heritage	Δ
	Creativity and innovativeness	Cultural	Enhancement of creativity and innovativeness as a result of the adaptive reuse	0
Generative capacity	Jobs creation	Economic	Creation of long-term jobs, directly and indirectly linked to the adaptive reuse	О
	Economic spillovers	Economic	Indirect and induced economic impacts in the area incl. diverse sectors such as building construction, cultural and creative activities, tourism, education and training, technologies for sustainability, research and innovation, entrepreneurship, etc.	0

Table 1. Cont

Circularity Dimension	Criteria	Sustainability Dimension	Description/Motivation	Comparison with Current Indicators
	Public finance benefit	Economic	Avoided costs for public finance and direct and indirect benefits from adaptive reuse interventions	Δ
	Attractiveness for creative, cultural and innovative enterprises	Economic	Localisation of innovative entrepreneurs, cultural and creative industries, research and development activities	Δ
	Attractiveness for residents	Economic	Localisation of permanent or temporary residents as a result of the adaptive reuse	Δ
	Attractiveness for circular cultural tourism	Economic	Localisation of economic activities linked to circular cultural tourism and hospitality as a result of the adaptive reuse	Δ
	Soil consumption reduction	Environmental	Avoided natural and fertile soil consumed, incl. for new buildings construction	0
	Air quality and microclimate	Environmental	Contribution to air quality and microclimate quality in the heritage context area	0
	GHG emissions reduction	Environmental	Contribution to GHG emissions reduction, incl. embodied energy valorisation of buildings and sites	0
	Water quality	Environmental	Contribution to enhancement of water quality in urban and rural environment	0
	Biodiversity	Environmental	Contribution to biodiversity conservation and enhancement, incl. actions to halt and reverse biodiversity loss	0
	Landscape quality and atmosphere	Social	Contribution to beauty and harmony of the landscape, enhancement of "place atmosphere", place-making	Δ
	Safety of public spaces	Social	Contribution to safety and accessibility of public spaces "for all"	Δ
	Cleanliness and healthiness of public spaces	Social	Contribution to cleanliness and healthiness of public spaces	Δ
	Quality of life for residents	Social	Contribution to objective elements of quality of life of residents, incl. presence of proximity shops, avoidance of gentrification effects and "touristification" of heritage sites	Δ
	Health	Social	Contribution to health incl. healthy materials, green installations, indoor air quality, natural lighting, noise control, electromagnetic pollution, healthy productions (e.g., healthy food in rural landscapes), mental health	Х
	Wellbeing	Social	Enhancement to self-perceived wellbeing for citizens and users, related to the adaptive reuse intervention	Х

Table 1. Cont

Source: Authors' elaboration, based on the work conducted under Horizon 2020 CLIC project, Work Package 2 on "Creating evidence base of cultural heritage impacts", co-developed with Horizon 2020 WP2 partners as described in the Acknowledgments section.

Figure 5 visualises circularity criteria structured into three groups: "Resources" linked to the re-generative capacity. "Circularity enabling factors" linked to the symbiotic capacity in the context area and "Outcomes" linked to the generative capacity of the heritage system.

Circular Cultural Heritage Adaptive Reuse



Figure 5. Visualisation of Circular CHAR database of criteria (Source: adapted from "Tripod model" by Luigi Fusco Girard, Horizon 2020 CLIC project).

4. Discussion

The performed literature structured review and meta-analysis has demonstrated that still many different interpretations of heritage indicators exist and there is still a lack of agreement in the scientific community to achieve a more uniform vision. Results showed that while some indicators for cultural heritage impacts assessment in the perspective of the circular economy are available, many relevant circularity aspects are not considered in current studies on cultural heritage impacts. Different indicators with a ranging degree of connections to cultural heritage have been developed over the years with a gaining attention especially in the most recent period. Yet, considerable effort is still necessary in order to reach a shared and harmonised vision on which is the most appropriate set of indicators to be used to assess the impact of cultural heritage adaptive reuse in a circular perspective.

From the point of view of the methodological approaches identified in the diverse sources analysed, a general confusion between criteria and indicators was observed, due to the lack of reference to a reliable and scientifically valid definition of indicators and how they should be used in heritage impacts research. The confusion present at the theoretical level is reflected at the practical level in the lack of a clearly defined and comprehensible structure that follows the methodological rigour of defining objectives, criteria and indicators. The result is a high variability of approaches in which, in the face of a declared and common need to identify an evaluation framework for heritage impacts assessment, each study approaches the subject from a diverse perspective. The lack of a clear methodology also impedes its replicability and could lead to slow down progress in this field.

This study proposed a structured review integrated with a meta-analysis of different kinds of sources, selected on the basis of the presence of indicators directly or indirectly linked to cultural heritage. More specifically, we proposed a classification of the sources according to the typology and the impact sectors and our findings show that one of the most used typologies of sources is the Institutional Report on cultural and creative sectors, while the sources directly dealing with cultural heritage ranked second. Furthermore, the temporal frequency graph shows that attention to these issues is growing rapidly and it has undergone a significant increase especially in recent years. Secondly, we proposed a classification of the indicators based on the dimension, scale, evaluation phase and typology. Our findings show that most of the proposed ones are quantitative indicators belonging to the social dimension, referring to the macro scale and allowing to perform an ex-post evaluation. From the literature review conducted, it is evident that there is a lack of a defined and shared methodology for the evaluation of the circular economy. This gap does not only concern an operational aspect, but rather originates from a sectorial "pillar" approach that hinders the reading of the relations and connections existing between different aspects that characterise the complex reality in which we live.

It is clear that, although the idea that cultural heritage contains both tangible and intangible values is now widely accepted, hard values, which are tangible and therefore measurable, are dominant because in decision-making processes they offer the possibility to choose on the basis of objective data. This means that soft, intangible and non-monetisable values are still neglected and this tendency was confirmed in our analysis by the predominance of quantitative over qualitative indicators. Proposing a multidimensional assessment of the impacts of the adaptive reuse of cultural heritage from the perspective of the circular economy model implies adopting a systemic perspective that is embedded in this model, capable of considering connections and interrelationships that exist between different systems. Despite the undeniable complexity, the evaluation tool can be interpreted as a means for simplification. It would not only orient the evaluation process towards an integration of the different values but would also favour a better management of the criticalities that the decision-making process itself poses at each stage of its development.

Analysing the specific scopes and domains of application of heritage indicators, the sources belonging to scopes "other than cultural heritage" have been taken into account to broaden the view and consider all the possible multidimensional impacts connected with the adaptive reuse. Among them, it can be highlighted that the cultural and creative sector domain includes specific indicators related to cultural heritage and to its potential impacts [25,67–69]. Another category of indicators that can be transferred to the cultural heritage sphere comes from the tourism sector, with a view on sustainability and culture. Indeed, Vecco and Srakar [83], while dealing with the difficult issue of building a sustainability index for cultural heritage, discovered that the biggest effort is generally dedicated to the relationship between sustainability and cultural heritage tourism. Among the additional analysed papers, many focus on sustainability in tourism, for example Asmelash and Kumar [84] try to link sustainable heritage tourism with tourist satisfaction, while Blancas et al. [85] deal with the construction of composite indicators for the sustainability assessment of a destination, including heritage-related indicators [85]. Moreover, Choi and Sirakaya [86] try to measure community tourism development, establishing a set of indicators according to five different dimensions (economic, social, cultural, ecological, political, and technological) and in the same way Lozano-Oyola et al. [87] build a sustainable tourism indicator system for cultural destinations. In this context, it is also worth mentioning the analysis of Ngamsomsuke et al. [88], who develop a set of indicators for sustainable cultural heritage development.

A further important correlation, when we deal with adaptive reuse as a means to implement circular economy, is represented by sustainability and circular economy in the built environment. Regenerating abandoned cultural heritage assets and landscapes determines the reduction of land consumption and the preservation of ecosystem services as well as the reduction of Construction and Demolition Waste (CDW) and the valorisation of embodied energy, i.e., the energy required in all the life cycle phases linked to the creation of the building [47]. Consequently, there is an urgent need of developing

environmental indicators able to quantitatively demonstrate how adaptive reuse has positive repercussions on the environment and reduces the input and output flows, from an Urban Metabolism (UM) perspective. Some initial studies focus on the benefits of adaptive reuse, underlining the significant reductions in energy consumption and in the emission of environmental carbon dioxide and other greenhouse gas, fossil fuel consumption, fresh water consumption, and materials use [42]. Foster and Krenin review the state of the art of the environmental indicators used for cultural heritage adaptive reuse [43]. From their analysis, it is clear that there are already many connections between the circular economy and the built environment and especially between cultural heritage adaptive reuse and the subsequent environmental advantages. As a matter of fact, "CE means using what is already there to maximise the use of embodied energy and materials in existing building stock. The challenge is that existing building stock, including cultural heritage buildings, must be refurbished and reused to meet the goals of a low-carbon economy" [43]. Despite this, the results of their research also underline that environmental indicators are rarely applied in cultural heritage adaptive reuse projects, pointing out the gap that exists between the indicators commonly used when we deal about circularity and the use of more specific indicators aimed at demonstrating the environmental advantages of cultural heritage adaptive reuse applications.

Finally, according to the results of the study conducted, interesting emerging fields of research intersecting circular economy and cultural heritage adaptive reuse are related to Intrinsic value, Local circular economies, Heritage community, Circular metabolism, Smart Specialisation Strategies, Wellbeing and Health impacts. For example, the concept of Intrinsic value of cultural heritage was proposed by Fusco Girard in 1987 [6] and further explored by Fusco Girard and Nijkamp in 1997 [8], but only recently the concept was developed by Fusco Girard and Vecco applying it to cultural heritage adaptive reuse [89,90], while operational tools and specific indicators for its assessment are still to be developed. On the same line, Gustafsson and Stanojev developed research on Smart Specialisation Strategies for cultural heritage adaptive reuse, pointing out the need of directing regional investments to achieve inter-sectorial synergic outcomes including cultural heritage adaptive reuse [91,92]. Moreover, the heritage community is another aspect related to a circular cultural heritage adaptive reuse as resulting from this study. However, indicators for assessing the role and contribution of the heritage community to cultural heritage adaptive reuse are still in their infancy and only few attempts have been made to structure robust evaluations [93]. Another important sector that can be in some way significant to associate to the adaptive reuse of cultural heritage is that related to wellbeing and health. Generally, there is not a direct correlation between cultural indicators and Societal Well-Being (SWB), with the exception of the Italian BES (equitable and sustainable well-being), while only few authors addressed the relationships between human health and heritage regeneration [94]. Therefore, considerable effort is still necessary to create this kind of correlation.

5. Conclusions

This paper provided a structured review and meta-analysis of literature sources on heritage impacts indicators, defining an impact assessment framework for cultural heritage adaptive reuse in the perspective of the emergent circular economy and circular city model (Circular CHAR impacts assessment framework). A classification of the selected sources and the indicators was performed, according to simple categorisations. The novelty of this paper consists in the attempt to link the concepts of cultural heritage, adaptive reuse and circular economy in an overall comprehensive impacts assessment framework that takes into account an enlarged definition of circular economy from a "human-centred" cultural perspective. This approach has been proposed within the Horizon 2020 CLIC research project and adopted as key reference for this study.

The main challenge with respect to the identification of a comprehensive impact assessment framework for circular adaptive reuse of cultural heritage is the definition of a common and recognised framework of criteria and indicators to assess quantitatively and qualitatively the multidimensional impacts of cultural heritage adaptive reuse projects and their ability to "close the loops" of diverse resources flows in line with the circular economy principles. The analysis conducted in this paper allowed to build a first and preliminary matrix of Circularity dimensions and criteria for Circular CHAR, open to further developments. The "Circular CHAR" impact assessment framework was structured for ex-post evaluation. Assuming the systemic perspective of the circular economy model implies that the evaluation phase should not be limited only to the ex-post assessment, taking into account the potential of using a structured framework of criteria and indicators also in the ex-ante and monitoring stages of the CHAR process, supporting decisionmaking and strategic planning processes and allowing to intervene in a preventive way to orient choices towards sustainability. Therefore, the Circular CHAR developed in this paper can be an excellent monitoring tool for circularity that allows coherent intervention on cultural heritage.

Circularity criteria can be adopted for ex-ante evaluation linked to decision-support systems, however cautious attention should be put to adapt the indicators and evaluation methods, since some of the impacts may be hard to be estimated and would need the identification of proxies to be used within multi-criteria analysis methods. Additionally, in the case of ex-ante evaluation (decision-support oriented), the involvement of multiple stakeholders and "points of view" could be key for effective and shared decisions in the longer term. It is important to highlight that the number of criteria identified for a full circularity assessment in ex-post evaluation can be hardly used in ex-ante decision-support evaluations. As decisions require the involvement of decision-makers, it is important to ensure that a limited number of relevant criteria are identified. For ex-ante evaluations, the relevant criteria can be selected according to the scale, stage and data availability of indicators. A hierarchy of priority for criteria and indicators can be identified applying diverse multi-criteria decision aiding approaches. The exercise of reducing and prioritising criteria and indicators in ex-ante evaluation processes would benefit the effectiveness of decisions, including the point of view of multiple actors and interests. In this way, indicators can become a tool for discussion and sharing between community members, facilitated by open and deliberative laboratories such as policy labs [95,96].

Finally, there must be no time lag between the transition to a "circular" and "humancentred" model of sustainable development and the elaboration and implementation of multidimensional assessment processes: these two perspectives should be integrated from the outset into ongoing evaluation processes in order to bring about a real change in the definition of strategies and actions at both local, national and international level.

The possible limitations of the study can be linked to subjective interpretations of indicators that was noticed in the first phase of the analysis, establishing ad hoc rules in order to position correctly the indicators within the categorisation table. We see therefore a need for in-depth studies aimed at improving the definition of indicator and at proposing well-established matrixes, with a clear distinction between quantitative and qualitative variables. Indeed, one of the greatest difficulties encountered while conducting the study consisted in identifying the typology, scale and sustainability dimension of the indicators. For example, depending on the context, indicators can be placed in the economic rather than cultural or social dimension. Sometimes there is a very thin and blurred boundary between these dimensions, and it would be appropriate to present more exhaustive descriptions and clearly structured matrices in future analyses, in which all the attributes that characterise the proposed indicators are shown. With regard to the environmental dimension, this is more easily recognisable, but at the same time very rare and only few studies addressing the environmental advantages of CHAR have appeared in the international literature. This opens the ground for new analyses linked to the environmental implications of cultural heritage adaptive reuse at different scales, therefore not only the single building, but also the possible impacts on the circular city metabolic functioning.

Finally, it is worth to note that the contribution of cultural heritage to the implementation of the European Green Deal [97] has been recently highlighted by ICOMOS and Europa Nostra to build an internationally agreed "cultural heritage green policy agenda" through the European Cultural Heritage Green Paper [81]. The present study aimed to contribute to the evolving international policies scenario for CHAR by providing a comprehensive framework that could be used at different levels for planning and design, as well as for the assessment of planned, ongoing and realised CHAR projects.

Supplementary Materials: The following are available online at https://www.mdpi.com/article/10 .3390/su13094759/s1: Table S1: Literature sources analysed.

Author Contributions: Conceptualisation, A.G. and L.F.G.; methodology, A.G.; formal analysis, S.I. and M.B.; investigation, S.I., M.B., A.G. and P.D.T.; resources, S.I. and M.B.; data curation, S.I. and M.B.; writing—original draft preparation, A.G., S.I. and M.B.; writing—review and editing, A.G., S.I. and M.B.; visualisation, A.G. and S.I.; supervision, A.G., P.D.T. and L.F.G.; project administration, L.F.G. and A.G. and funding acquisition, L.F.G. and A.G. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded under the framework of Horizon 2020 research project CLIC: Circular models Leveraging Investments in Cultural heritage adaptive reuse. This project received funding from the European Union's Horizon 2020 research and innovation program under Grant Agreement No 776758.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: No new data were created or analysed in this study. Data sharing is not applicable to this article.

Acknowledgments: The authors gratefully acknowledge the contribution of the Horizon 2020 CLIC project researchers and additional experts who agreed to be involved in this study. We are particularly thankful for the discussions, scientific literature shared, research reports, previous joint publications and thoughtful comments received.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses or interpretation of data; in the writing of the manuscript or in the decision to publish the results.

Appendix A

Table A1. Literature sources on cultural heritage impact sectors and indicators. Ec: Economic indicators, S: Social Indicators,En: Environmental Indicators, C: Cultural Indicators.

Source	Year	Source Typology	Number of Indicators		Dimension	S		Scale	Phase	Typology	
				Ec	S	En	С			Quant	Qualit
Nijkamp [55]	1989	Article	6	3			3	Meso	Ex-post	3	3
Pearson et al. [98]	1998	Institutional Report	139	19	9	36	75	Meso	Ex-post	137	2
Greffe [99]	2004	Article	4	4				Meso	Ex-post	4	
Hockings et al. [63]	2008	Institutional Report	17	2	7	5	3	Meso	Ex-post	17	
Labadi [59]	2011	Institutional Report	57	24	22	7	4	Meso	Ex-post	35	22
Rypkema and Cheong [60]	2011	Article	29	16	17	6		Meso	Ex-post	8	21
Licciardi and Amirtahmasebi [5]	2011	Institutional Report	13	4	7	2		Meso	Ex-post	7	6
Zancheti and Hidaka [100]	2011	Article	3				3	Meso	Ex-post		3

Source	Year	Source Typology	Number of Indicators	Dimensions				Scale	Phase	Typology	
				Ec	S	En	С			Quant	Qualit
Elsorady [53]	2014	Article	16	1	3	3	9	Micro	Ex-ante		16
CHCfE Consortium [56]	2015	Research Report	54	18	13	9	14	Meso	Ex-post	32	22
Fusco Girard et al. 2015 [49]	2015	Article	124	77	14	15	18	Meso	Ex-post	121	3
James [101]	2015	Article	40	8	16	9	7	Meso	Ex-post	35	22
Sowińska-Świerkosz [50]	2017	Article	15	3	1	7	4	Meso	Ex-post	10	5
Guzmán et al. [52]	2017	Article	14		4	6	4	Meso	Ex-post	12	2
Nocca [48]	2017	Article	178	111	35	1	31	Meso	Ex-post	172	6
Stanik et al. [41]	2018	Article	6			2	4	Macro	Ex-post	3	3
Vecco and Srakar [83]	2018	Article	7	2	1	1	3	Macro	Ex-post	7	
Airaghi et al. [61]	2019	Research Report	13	12	1			Meso	Ex-post	13	
Historic England [102]	2019	Research Report	41	7	5	7	22	Macro	Ex-post	41	
Günçe and Mısırlısoy [17]	2019	Article	25	5	4	6	10	Micro	Ex-post		25
Della Spina [16]	2020	Article	11	6		4	1	Micro	Ex-ante	4	7
De Leão Dornelles et al [103]	2020	Article	1				1		Ex-post		1
Melloni et al. [104]	2020	Research Report	144	19	82	7	36	Meso	Ex-post	121	23

Table A1. Cont.

Appendix B

Table A2. Literature sources on other impact sectors and indicators. Ec: Economic indicators, S: Social Indicators, En: Environmental Indicators, C: Cultural Indicators.

Source	Year	Source Typology	Number of Indicators	1	Dimension	5		Scale	Phase	Typology	
				Ec	S	En	С			Quant	Qualit
Mercer [105]	2002	Institutional Report	373	56	204	21	92	Macro	Ex-post	181	192
WTO [67]	2004	Institutional Report	29	6	7	16		Macro	Ex-post	24	5
OCPA Task Force [106]		Institutional Report	64	15	21	2	26	Macro	Monitoring	21	43
Home Affairs Bureau [107]	2005	Institutional Report	27	7	10		10	Macro	Ex-post	12	15
Choi and Sirakaya [86]	2006	Article	98	16	42	30	10	Meso	Ex-post	35	63
OECD [108]	2006	Institutional Report	104	68	26		10	Macro	Ex-post	103	1
Jackson et al. [109]	2006	Research Report	53	16	9	1	27	Macro	Ex-post	46	7
UNESCO [110]	2007	Institutional Report	21	3	3		15	Meso	Ex-post	19	2

Source	Year	Source Typology	Number of Indicators		Dimension	s		Scale	Phase	Турс	ology
				Ec	S	En	С			Quant	Qualit
UNESCO [111]	2007	Institutional Report	23	11	8		4	Macro	Ex-post	23	
Institut de la statistique du Québec [112]	2007	Institutional Report	67	29	29		9	Macro	Ex-post	56	11
OECD [113]	2008	Institutional Report	31	20	3	1	7	Macro	Ex-post	21	10
Ministry for Culture and Heritage [114]	2009	Institutional Report	17	5	5		7	Macro	Ex-post	15	2
KEA European Affairs [115]	2009	Institutional Report	11	3	5		3	Macro	Ex-post	11	
NCCRS [116]	2010	Institutional Report	16	6	4		6	Macro	Ex-post	16	
UNESCO [117]	2010	Institutional Report	312	83	99	15	115	Macro	Ex-post	177	135
Ngamsomsuke et al. [88]	2011	Article	20	4	3	7	6	Macro	Ex-post		20
Daschko [118]	2011	Institutional Report	47	5	28	2	12	Macro	Ex-post	7	40
Ministry of Culture and Education of Finland [119]	2011	Institutional Report	116	54	33	3	26	Macro	Ex-post	105	11
ESSnet-CULTUR [120]	2012	Research Report	28	9	11		8	Macro	Ex-post	28	
Lozano Oyola et al. [85]	2012	Article	62	20	6	28	8	Meso	Ex-post	54	8
Montalto [121]	2012	Institutional Report	23	13	5	1	4	Meso	Ex-post	16	7
UNESCO and UNDP [122]	2013	Research Report	54	23	18	2	11	Macro	Ex-post	44	10
Columbia Basin Rural Development Institute [123]	2013	Research Report	164	48	36	6	74	Macro	Ex-post	94	70
Oxford Economics [124]	2013	Research Report	4	4				Macro	Ex-post	4	
UCLG [125]	2014	Institutional Report	57		34	5	18	Meso	Ex-post		57
ARTS COUNCIL ENGLAND [126]	2014	Institutional Report	59		51		8	Micro	Ex-post		59
UNESCO [127]	2014	Institutional Report	22	2	13		7	Macro	Ex-post	16	6
United Nations [128]	2015	Institutional Report	53	12	27	8	6	Macro	Ex-post	53	
Global network "Future we want includes culture" [129]	2015	Institutional Report	28	2	8	5	13	Macro	Ex-post	26	2
Kushner and Cohen [130]	2016	Research Report	17	15	2			Macro	Ex-post	28	
Council of Europe [131]	2016	Institutional Report	26	5	14		7	Macro	Ex-post	16	10
University of Baltimore [132]	2016	Institutional Report	56	6	27	18	5	Meso	Ex-post	53	3

Table A2. Cont.

Source	Year	Source Typology	Number of Indicators	Dimensions				Scale	Phase	Typology	
				Ec	S	En	С			Quant	Qualit
Blancas et al. [85]	2016	Article	52	26	11	12	3	Macro	Ex-post	52	
European Commission [133]	2016	Research Report	43	14	10	18	1	Meso	Ex-post	41	2
UNESCO [134]	2017	Institutional Report	60	6	20	13	21	Macro	Ex-post	36	24
Ortega-Villa and Ley-Garcia [135]	2017	Article	19		12		7	Meso	Ex-post	4	15
Ren and Han [136]	2018	Article	67	14	18	24	11	Micro	Ex-post	41	26
ISTAT [137]	2018	Research Report	2	2				Macro	Ex-post	2	
EUROSTAT [138]	2019	Research Report	14	3	2		9	Macro	Ex-post	12	2
EUROSTAT [139]	2019	Research Report	9	1	4	4		Macro	Ex-post	9	
UNESCO [140]	2019	Research Report	21	7	6	1	7	Macro	Ex-post	1	8
Montalto et al. [141]	2019	Article	12	4	4		4	Meso	Ex-post	7	5
ISTAT [142]	2018	Research Report	11	2	3	6		Macro	Ex-post	9	2
Asmelash and Kumar [84]	2019	Article	61	10	31	12	8	Macro	Ex-post	14	47
European Commission [143]	2019	Website	16	8	2	6		Macro	Ex-post	14	2
European Commission [144]	2019	Institutional Report	29	3	21		5	Meso	Ex-post	23	6
OECD and ICOM [145]	2019	Institutional Report	8	2	6			Macro	Monitoring	2	6
Compendium of Cultural Policies & Trends [72]	2019	Website	26	8	11		7	Macro	Ex-post	18	8

Table A2. Cont.

Appendix C

Table A3. Literature sources on Circular Economy impact sectors and indicators. Ec: Economic indicators, S: Social Indicators, En: Environmental Indicators, C: Cultural Indicators.

Source	Year	Source Typology	Number of Indicators	Dimensions				Scale	Phase	Typology	
				Ec	S	En	С			Quant	Qualit
Gravagnuolo et al. [36]	2019	Article	17			17		Meso	Ex-post	17	
Historic England [146]	2019	Research Report	2			2		Micro	Ex-post	2	
Foster and Kreinin [43]	2020	Article	12			12		Micro	Ex-post	12	
Foster et al. [44]	2020	Article	20			20		Micro	Ex-post	20	
Heisel and Rau-Oberhuber [147]	2020	Article	3			3		Micro	Ex-post	3	

References

- 1. European Commission. Towards an Integrated Approach to Cultural Heritage for Europe; EUR-Lex: Bruxelles, Belgium, 2014.
- 2. Wijkman, A.; Skånberg, K. The Circular Economy and Benefits for Society: Jobs and Climate Clear Winners in an Economy Based on Renewable Energy and Resource Efficiency; Club Rome: Rome, Italy, 2015.
- 3. Bokova, I.G. UNESCO Director-General 2009–2017 writer of foreword. In *Culture: Urban Future Global Report on Culture for Sustainable Urban Development;* UNESCO: Paris, France, 2016; ISBN 978-92-3-100170-3.
- 4. Australia ICOMOS. *The Burra Charter: The Australia ICOMOS Charter for Places of Cultural Significance;* Australia/ICOMOS: Burwood, Australia, 2013.
- Licciardi, G.; Amirtahmasebi, R. The Economics of Uniqueness. Lnvesting in Historic City Cores and Cultural Heritage Assets for Sustainable Development; The World Bank: Washington, DC, USA, 2012; ISBN 978-0-8213-9650-6.
- 6. Fusco Girard, L. Risorse Architettoniche e Culturali: Valutazioni e Strategie di Conservazione; Franco Angeli: Milan, Italy, 1987.
- Fusco Girard, L.; Nijkamp, P. Energia, Bellezza, Partecipazione: La Sfida Della Sostenibilità. Valutazioni Integrate tra Conservazione e Sviluppo; Franco Angeli: Milan, Italy, 2004; ISBN 8846462521.
- 8. Fusco Girard, L.; Nijkamp, P. Le Valutazioni per lo Sviluppo Sostenibile Della Città e del Territorio; Fusco Girard, L., Nijkamp, P., Eds.; FrancoAngeli: Milan, Italy, 1997; ISBN 8846401824.
- 9. Ost, C.; Carpentier, F. Conservation Dynamics: A New Paradigm for Cultural Heritage in Times of Economic Changes. In *Innovative Built Heritage Models*; Van Balen, K., Vandesande, A., Eds.; CRC Press: Boca Raton, FL, USA, 2018.
- 10. Bullen, P.A.; Love, P.E.D. Adaptive reuse of heritage buildings. Struct. Surv. 2011, 29, 411–421. [CrossRef]
- 11. Fusco Girard, L. Implementing the circular economy: The role of cultural heritage as the entry point. Which evaluation approaches? *BDC Boll. Del Cent. Calza Bini* 2019, *19*, 245–277. [CrossRef]
- 12. Gravagnuolo, A.; Fusco Girard, L.; Ost, C.; Saleh, R. Evaluation criteria for a circular adaptive reuse of cultural heritage. *BDC Boll. Del Cent. Calza Bini* **2017**, *17*, 185–216.
- 13. Gustafsson, C. Conservation 3.0—Cultural heritage as a driver for regional growth. Sci. Res. Inf. Technol. 2019, 9, 21–32. [CrossRef]
- 14. Douglas, J. Building Adaptation, 2nd ed.; Routledge: London, UK, 2006; ISBN 13:978-0-7506-6667-1.
- 15. De Medici, S.; De Toro, P.; Nocca, F. Cultural heritage and sustainable development: Impact assessment of two adaptive reuse projects in Siracusa, Sicily. *Sustainability* **2020**, *12*, 311. [CrossRef]
- 16. Della Spina, L. Adaptive sustainable reuse for cultural heritage: A multiple criteria decision aiding approach supporting urban development processes. *Sustainability* **2020**, *12*, 1363. [CrossRef]
- 17. Günçe, K.; Misirlisoy, D. Assessment of adaptive reuse practices through user experiences: Traditional houses in thewalled city of Nicosia. *Sustainability* **2019**, *11*, 540. [CrossRef]
- 18. Rayman, M.; Boyle, R.; Yang, A.Y.; Tangari, J. Adaptive reuse: A review and analysis of its relationship to the 3 Es of sustainability. *Facilities* **2017**, *35*, 138–154. [CrossRef]
- 19. Partnership on Circular Economy and Sustainable Land Use, Handbook—Sustainable and Circular Re-Use of Spaces and Buildings; European Commission: Brussels, Belgium, 2019; pp. 1–72.
- 20. Angrisano, M.; Fusco Girard, L.; Bianchi, A. A literature review about life cycle assessment as a tool to support circular economy innovation in the built environment sector. *BDC Boll. Del Cent. Calza Bini* **2019**, *19*, 125–143. [CrossRef]
- Geissdoerfer, M.; Savaget, P.; Bocken, N.M.P.; Hultink, E.J. The Circular Economy—A new sustainability paradigm? J. Clean. Prod. 2017, 143, 757–768. [CrossRef]
- 22. Kirchherr, J.; Reike, D.; Hekkert, M. Conceptualizing the circular economy: An analysis of 114 definitions. *Resour. Conserv. Recycl.* 2017, 127, 221–232. [CrossRef]
- Ghisellini, P.; Cialani, C.; Ulgiati, S. A review on circular economy: The expected transition to a balanced interplay of environmental and economic systems. J. Clean. Prod. 2016, 114, 11–32. [CrossRef]
- 24. Moraga, G.; Huysveld, S.; Mathieux, F.; Blengini, G.A.; Alaerts, L.; Van Acker, K.; de Meester, S.; Dewulf, J. Circular economy indicators: What do they measure? *Resour. Conserv. Recycl.* 2019, 146, 452–461. [CrossRef]
- 25. Parchomenko, A.; Nelen, D.; Gillabel, J.; Rechberger, H. Measuring the circular economy—A Multiple Correspondence Analysis of 63 metrics. *J. Clean. Prod.* 2019, 210, 200–216. [CrossRef]
- Saidani, M.; Yannou, B.; Leroy, Y.; Cluzel, F.; Kendall, A.; Saidani, M.; Yannou, B.; Leroy, Y.; Cluzel, F.; Kendall, A. A taxonomy of circular economy indicators To cite this version: HAL Id: Hal-01954800. J. Clean. Prod. 2019, 207, 542–559. [CrossRef]
- 27. Lemille, A. Optimizing Circular Value; Circular Economy Club: Boston, MA, USA, 2017.
- Moreau, V.; Sahakian, M.; van Griethuysen, P.; Vuille, F. Coming Full Circle: Why Social and Institutional Dimensions Matter for the Circular Economy. J. Ind. Ecol. 2017, 21, 497–506. [CrossRef]
- 29. World Health Organization Regional Office for Europe. *Circular Economy and Health: Opportunities and Risks;* World Health Organization Regional Office for Europe: Copenhagen, Denmark, 2018; ISBN 9789289053341.
- Watts, N.; Amann, M.; Arnell, N.; Ayeb-Karlsson, S.; Belesova, K.; Berry, H.; Bouley, T.; Boykoff, M.; Byass, P.; Cai, W.; et al. The 2018 report of the Lancet Countdown on health and climate change: Shaping the health of nations for centuries to come. *Lancet* 2018, 392, 2479–2514. [CrossRef]
- 31. Gupta, J.; Hurley, F.; Grobicki, A.; Keating, T.; Stoett, P.; Baker, E.; Guhl, A.; Davies, J.; Ekins, P. Communicating the health of the planet and its links to human health. *Lancet Planet. Health* **2019**, *3*, e204–e206. [CrossRef]

- Watts, N.; Amann, M.; Ayeb-Karlsson, S.; Belesova, K.; Bouley, T.; Boykoff, M.; Byass, P.; Cai, W.; Campbell-Lendrum, D.; Chambers, J.; et al. The Lancet Countdown on health and climate change: From 25 years of inaction to a global transformation for public health. *Lancet* 2018, 391, 581–630. [CrossRef]
- Watts, N.; Amann, M.; Arnell, N.; Ayeb-Karlsson, S.; Beagley, J.; Belesova, K.; Boykoff, M.; Byass, P.; Cai, W.; Campbell-Lendrum, D.; et al. The 2020 report of The Lancet Countdown on health and climate change: Responding to converging crises. *Lancet* 2021. [CrossRef]
- 34. ARUP. Circular Economy in the Built Environment; ARUP: London, UK, 2016.
- 35. Ellen MacArthur Foundation. ARUP Circular Economy in Cities: Project Guide; Ellen MacArthur Foundation: London, UK, 2019.
- 36. Gravagnuolo, A.; Angrisano, M.; Girard, L.F. Circular economy strategies in eight historic port cities: Criteria and indicators towards a circular city assessment framework. *Sustainability* **2019**, *11*, 3512. [CrossRef]
- 37. Fusco Girard, L.; Nocca, F.; Gravagnuolo, A. Matera: City of nature, city of culture, city of regeneration. Towards a landscapebased and culture-based urban circular economy. *Aestimum* **2019**, *74*, 5–42. [CrossRef]
- 38. Costanza, R.; de Groot, R.; Sutton, P.; van der Ploeg, S.; Anderson, S.J.; Kubiszewski, I.; Farber, S.; Turner, R.K. Changes in the global value of ecosystem services. *Glob. Environ. Chang.* **2014**, *26*, 152–158. [CrossRef]
- Alcamo, J.; Ash, N.J.; Butler, C.D.; Callicott, J.B.; Capistrano, D.; Carpenter, S.R.; Castilla, J.C.; Chambers, R.; Chopra, K.; Cropper, A.; et al. *Ecosystems and Human Well-being: A Framework for Assessment*; Island Press: Washington, DC, USA, 2003; ISBN 1-55963-403-0.
- 40. TEEB. *The Economics of Ecosystems and Biodiversity Ecological and Economic Foundations*; Kumar, P., Ed.; Earthscan: London, UK; Washington, DC, USA, 2010.
- 41. Stanik, N.; Aalders, I.; Miller, D. Towards an indicator-based assessment of cultural heritage as a cultural ecosystem service—A case study of Scottish landscapes. *Ecol. Indic.* 2018, *95*, 288–297. [CrossRef]
- 42. Foster, G. Circular economy strategies for adaptive reuse of cultural heritage buildings to reduce environmental impacts. *Resour. Conserv. Recycl.* 2020, 152, 104507. [CrossRef]
- 43. Foster, G.; Kreinin, H. A review of environmental impact indicators of cultural heritage buildings: A circular economy perspective. *Environ. Res. Lett.* **2020**, *15*, 043003. [CrossRef]
- 44. Foster, G.; Kreinin, H.; Stagl, S. The future of circular environmental impact indicators for cultural heritage buildings in Europe. *Environ. Sci. Eur.* **2020**, *32*, 1–17. [CrossRef]
- 45. Cerreta, M.; De Toro, P. Urbanization suitability maps: A dynamic spatial decision support system for sustainable land use. *Earth Syst. Dyn.* **2012**, *3*, 157–171. [CrossRef]
- 46. Mendas, A.; Delali, A. Integration of MultiCriteria Decision Analysis in GIS to develop land suitability for agriculture: Application to durum wheat cultivation in the region of Mleta in Algeria. *Comput. Electron. Agric.* **2012**, *83*, 117–126. [CrossRef]
- Fusco Girard, L.; Gravagnuolo, A. Circular economy and cultural heritage/landscape regeneration. Circular business, financing and governance models for a competitive Europe. BDC Boll. Del Cent. Calza Bini 2017, 2017, 35–52.
- 48. Nocca, F. The role of cultural heritage in sustainable development: Multidimensional indicators as decision-making tool. *Sustainability* **2017**, *9*, 1882. [CrossRef]
- 49. Fusco Girard, L.; Gravagnuolo, A.; Nocca, F.; Angrisano, M.; Bosone, M. Towards an economic impact assessment framework for Historic Urban Landscape conservation and regeneration projects. *BDC Boll. Del Cent. Calza Bini* 2015, *15*, 1–29.
- Sowińska-Świerkosz, B. Review of cultural heritage indicators related to landscape: Types, categorisation schemes and their usefulness in quality assessment. *Ecol. Indic.* 2017, *81*, 526–542. [CrossRef]
- 51. OECD. Better Regulation Practices across the European Union. Chapter 4. Ex Post Review of Laws and Regulations Across the European Union; OECD: Paris, France, 2019.
- 52. Guzmán, P.C.; Roders, A.R.P.; Colenbrander, B.J.F. Measuring links between cultural heritage management and sustainable urban development: An overview of global monitoring tools. *Cities* **2017**, *60*, 192–201. [CrossRef]
- 53. Elsorady, D.A. Assessment of the compatibility of new uses for heritage buildings: The example of Alexandria National Museum, Alexandria, Egypt. J. Cult. Herit. 2014, 15, 511–521. [CrossRef]
- 54. Kutut, V.; Zavadskas, E.K.; Lazauskas, M. Assessment of priority alternatives for preservation of historic buildings using model based on ARAS and AHP methods. *Arch. Civ. Mech. Eng.* **2014**, *14*, 287–294. [CrossRef]
- 55. Nijkamp, P. Quantity and quality: Evaluation indicators for our cultural-architectural heritage. Ser. Res. Memo 1989, 46, 1–32.
- 56. CHCfE Consortium. Cultural Heritage Counts for Europe; CHCfE: Krakow, Poland, 2015.
- 57. Historic England. *HERITAGE COUNTS Heritage and the Economy* 2016; Historic England: London, UK, 2016. Available online: https://historicengland.org.uk/research/heritage-counts/heritage-and-economy/ (accessed on 16 April 2021).
- Historic England. HERITAGE COUNTS Heritage and Society 2016; Historic England: London, UK, 2016. Available online: https://historicengland.org.uk/research/heritage-counts/heritage-and-economy/ (accessed on 16 April 2021).
- 59. Labadi, S. *Evaluating the Socio-Economic Impacts of Selected Regenerated Heritage Sites in Europe;* European cultural foundation: Amsterdam, The Netherlands, 2011.
- 60. Rypkema, D.; Cheong, C. Measurements and indicators of heritage as development. In Proceedings of the ICOMOS 17th General Assembly, Paris, France, 27 November–2 December 2011; ICOMOS: Paris, France, 2011.

- 61. Lykogianni, E.; Mobilio, L.; Procee, R.; Airaghi, E.; Kern, P.; Le Gall, A.; Krohn, C.; Vanhoutte, C. *The MCH as a Strategic Territorial Development Resource: Mapping Impacts Through a Set of Common European Socio-Economic Indicators Target Analysis Interim Report Material Cultural Heritage as a Strategic Territorial Development Resource: Mapping Impacts Through a Set of Common European Socio-Economic Indicators. Synthesis Report; ESPON: Luxembourg, 2019.*
- 62. Heink, U.; Kowarik, I. What are indicators? On the definition of indicators in ecology and environmental planning. *Ecol. Indic.* **2010**, *10*, 584–593. [CrossRef]
- 63. Hockings, M.; James, R.; Stolton, S.; Dudley, N.; Mathur, V.; Makombo, J.; Courrau, J.; Parrish, J. Enhancing Our Heritage Toolkit Assessing Management Effectiveness of Natural World Heritage Sites; UNESCO World Heritage Centre: Paris, France, 2008.
- 64. Fry, G.; Tveit, M.S.; Ode, Å.; Velarde, M.D. The ecology of visual landscapes: Exploring the conceptual common ground of visual and ecological landscape indicators. *Ecol. Indic.* 2009, *9*, 933–947. [CrossRef]
- 65. Zhang, Y.; Yang, Z.; Li, W. Analyses of urban ecosystem based on information entropy. Ecol. Modell. 2006, 197, 1–12. [CrossRef]
- 66. European Environment Agency. *Europe's Environment the Dobris Assessment. Noise and Radiation;* Stanners, D.B.P., Ed.; EEA: Copenaghen, Denmark, 1995.
- 67. World Tourism Organization. Indicators of Sustainable Development for Tourism Destinations: A Guidebook; WTO: Madrid, Spain, 2004.
- 68. OECD-Organisation for Economic Co-operation and Development. *Measuring and Managing Results in Development Co-Operation;* OECD: Paris, France, 2014.
- 69. Mikolajewicz, N.; Komarova, S.V. Meta-Analytic Methodology for Basic Research: A Practical Guide. *Front. Physiol.* **2019**, *10*, 203. [CrossRef] [PubMed]
- Gravagnuolo, A.; Fusco Girard, L.; Kourtit, K.; Nijkamp, P. Adaptive re-use of urban cultural resources: Contours of circular city planning. *City Cult. Soc.* 2021. In press.
- 71. Fusco Girard, L. The circular economy in transforming a died heritage site into a living ecosystem, to be managed as a complex adaptive organism. *Aestimum* **2021**, 77, 145–180. [CrossRef]
- 72. Compendium of Cultural Policies & Trends The Compendium of Cultural Policies & Trends. Available online: https://www.culturalpolicies.net/ (accessed on 8 February 2021).
- EUROSTAT Circular Economy Indicators. Available online: https://ec.europa.eu/eurostat/web/circular-economy/indicators/ monitoring-framework (accessed on 8 February 2021).
- 74. Van Buren, N.; Demmers, M.; van der Heijden, R.; Witlox, F. Towards a Circular Economy: The Role of Dutch Logistics Industries and Governments. *Sustainability* **2016**, *8*, 647. [CrossRef]
- 75. Potting, J.; Hekkert, M.; Worrell, E.; Hanemaaijer, A. *Circular Economy: Measuring Innovation in the Product Chain*; PBL Netherlands Assessment Agency: The Hague, The Netherlands, 2017.
- 76. Ellen MacArthur Foundation. *Circularity Indicators. An Approach to Measuring Circularity. Project Overview;* Ellen MacArthur Foundation: Cowes, UK, 2015.
- 77. Ellen MacArthur Foundation. *Growth within: A Circular Economy Vision for a Competitive Europe;* Ellen MacArthur Foundation: Cowes, UK, 2015.
- 78. Ellen MacArthur Foundation. *Delivering the Circular Economy a toolkit for Policymakers;* Ellen MacArthur Foundation: Cowes, UK, 2015.
- 79. Bottero, M.; Lerda, M. Valutazione circolare degli interventi di riuso adattivo: Il caso della città di Torino. *BDC Boll. Del Cent. Calza Bini* **2019**, 19, 497–513. [CrossRef]
- 80. Fusco Girard, L.; Gravagnuolo, A. Il riuso del patrimonio culturale religioso: Criteri e strumenti di valutazione. *BDC Boll. del Cent. Calza Bini* 2018, 18, 237–246.
- 81. Potts, A. European Cultural Heritage Green Paper; Europa Nostra: The Hague, The Netherlands; Brussels, Belgium, 2021.
- Gravagnuolo, A.; Vellecco, I.; Fusco Girard, L.; Lauro, N.C.; Ikiz Kaya, D.; Pintossi, N.; Acri, M.; Dobricic, S.; Debevec, M.; Jokilehto, J.; et al. Survey on Best Practices of Cultural Heritage Adaptive Reuse (Horizon 2020 Deliverable D1.3). European Commission Horizon 2020 CLIC Deliverables: Naples, Italy, 2021.
- 83. Vecco, M.; Srakar, A. The unbearable sustainability of cultural heritage: An attempt to create an index of cultural heritage sustainability in conflict and war regions. *J. Cult. Herit.* **2018**, *33*, 293–302. [CrossRef]
- 84. Asmelash, A.G.; Kumar, S. The structural relationship between tourist satisfaction and sustainable heritage tourism development in Tigrai, Ethiopia. *Heliyon* **2019**, *5*, e01335. [CrossRef]
- 85. Blancas, F.J.; Lozano-Oyola, M.; González, M.; Caballero, R. Sustainable tourism composite indicators: A dynamic evaluation to manage changes in sustainability. J. Sustain. Tour. 2016, 24, 1403–1424. [CrossRef]
- 86. Choi, H.S.C.; Sirakaya, E. Sustainability indicators for managing community tourism. Tour. Manag. 2006, 27, 1274–1289. [CrossRef]
- 87. Lozano-Oyola, M.; Blancas, F.J.; González, M.; Caballero, R. Sustainable tourism indicators as planning tools in cultural destinations. *Ecol. Indic.* 2012, *18*, 659–675. [CrossRef]
- Ngamsomsuke, W.; Hwang, T.; Huang, C. Sustainable Cultural Heritage Tourism Indicators. In Proceedings of the International Conference on Social Science and Humanity IPEDR; 2011; Volume 5, pp. 516–520.
- 89. Fusco Girard, L.; Vecco, M. Genius Loci: The evaluation of places between instrumental and intrinsic values. *BDC Boll. Del Cent. Calza Bini* **2019**, 19, 473–495. [CrossRef]
- 90. Fusco Girard, L.; Vecco, M. The "Intrinsic Value" of Cultural Heritage as Driver for Circular Human-Centered Adaptive Reuse. *Sustainability* **2021**, *13*, 3231. [CrossRef]

- 91. Stanojev, J.; Gustafsson, C. Smart Specialisation Strategies for Elevating Integration of Cultural Heritage into Circular Economy. *Sustainability* 2021, *13*, 3685. [CrossRef]
- 92. Gustafsson, C.; Stanojev, J. *Economic Spillovers of Cultural Heritage Adaptive Reuse at Regional Level (Horizon 2020 CLIC Deliverable 2.1)*; European Commission Horizon 2020 CLIC Deliverables: Uppsala, Sweden, 2019.
- 93. Cerreta, M.; Giovene di Girasole, E. Towards Heritage Community Assessment: Indicators Proposal for the Self-Evaluation in Faro Convention Network Process. *Sustainability* **2020**, *12*, 9862. [CrossRef]
- 94. Carone, P.; De Toro, P.; Franciosa, A. Evaluation of Urban Processes on Health in Historic Urban Landscape Approach: Experimentation in the Metropolitan Area of Naples (Italy). *Qual. Innov. Prosper.* **2017**, *21*, 202. [CrossRef]
- 95. Olejniczak, K.; Borkowska-Waszak, S.; Domaradzka-Widła, A.; Park, Y. Policy labs: The next frontier of policy design and evaluation? *Policy Polit.* 2020, *48*, 89–110. [CrossRef]
- Olejniczak, K.; Borkowska-Waszak, S.; Domaradzka-Widla, A.; Park, Y. The Craft of Policy Labs: An Overview of Methods Applied for Development and Testing of Policy Solutions. Proceedings of 3rd International Conference on Public Policy (ICPP3), Singapore, 28–30 June 2017; p. 07041.
- 97. European Commission The European Green Deal. Eur. Comm. 2019. [CrossRef]
- Pearson, M.; Johnston, D.; Lennon, J.; McBryde, I.; Duncan, M.; Nash, D.; Wellington, B. Environmental Indicators for National State of the Environment Reporting. Natural and Cultural Heritage; State of the Environment: Canberra, Australia; Department of the Environment: Canberra, Australia, 1998.
- 99. Greffe, X. Is heritage an asset or a liability? J. Cult. Herit. 2004, 5, 301–309. [CrossRef]
- Zancheti, S.M.; Hidaka, L.T.F. Measuring urban heritage conservation: Theory and structure (part 1). J. Cult. Herit. Manag. Sustain. Dev. 2011, 1, 96–108. [CrossRef]
- 101. James, P. Urban Sustainability in Theory and Practice: Circles of sustainability; Routledge: London, UK, 2015; ISBN 9781315765747.
- 102. Historic England. *HERITAGE COUNTS Heritage Indicators* 2019; Historic England: London, UK, 2019. Available online: https://historicengland.org.uk/content/heritage-counts/pub/2019/hc2019-heritage-indicators/ (accessed on 22 April 2021).
- 103. De Leão Dornelles, L.; Gandolfi, F.; Mercader-Moyano, P.; Mosquera-Adell, E. Place and memory indicator: Methodology for the formulation of a qualitative indicator, named place and memory, with the intent of contributing to previous works of intervention and restoration of heritage spaces and buildings, in the aspect of sustainabi. *Sustain. Cities Soc.* **2020**, *54*, 101985. [CrossRef]
- 104. Espon. *HERIWELL—Cultural Heritage as a Source of Societal Well-being in European Regions. Inception Report—Conceptual framework;* Espon EGTC: Luxembourg, 2020.
- 105. Mercer, C. Towards Cultural Citizenship: Tools for Cultural Policy and Development; Gidlund: Stockholm, Sweden, 2012; ISBN 9178446228.
- 106. Task Force on Cultural Indicators of Human Development in Africa. Cultural Indicators: Views from Africa. 2004. Available online: https://ocpa.irmo.hr/activities/meetings/maputo2004/cultural_indicators-en.pdf (accessed on 14 April 2021).
- 107. Home Affairs Bureau The Hong Kong Special Administrative Region Government A study on Creativity Index. *A Study Creat. Index* **2005**, 1–100.
- 108. Gordon, J.C.; Beilby-Orrin, H. International Measurement of the Economic and Social Importance of Culture; OECD: Paris, France, 2006.
- 109. Jackson, M.R.; Kabwasa-Green, F.; Herranza, J. Cultural Vitality in Communities: Interpretation and Indicators; The Urban Institute: Washington, DC, USA, 2006.
- 110. van der Pol, H. Key Role of Cultural and Creative Industries in the Economy Introduction: Increasing Role of Cultural and Creative Industries; UNESCO Institute for Statistics: Montreal, QC, Canada, 2007.
- 111. UNESCO. Statistics on Cultural Industries; UNESCO: Bangkok, Thailand, 2007.
- 112. Observatoire De La Culture Et Des Communications Du Québec A System of Indicators for Culture and Communications in Quebec. Part One: Conception and Concerted. Development of the Indicators; Institut de la Statistique du Québec: Quebec, QC, Canada, 2007.
- 113. OECD. The Impact of Culture on Tourism; OECD Publishing: Paris, France, 2008; ISBN 9789264056480.
- 114. Ministry for Culture and Heritage. *Cultural Indicators for New Zealand;* Statistics New Zealand: Wellington, New Zealand; Ministry for Culture and Heritage: Wellington, New Zealand, 2009.
- 115. Affair, K.E.A. The Impact of Culture on Literacy. A Study Prepared for the European Commission; Europen Commission: Brussels, Beglgium, 2009.
- 116. Culture Ministers Council's Statistics Working Group—Australian Bureau of Statistics National Centre for Culture and Recreation Statistics. *Vital Signs: Cultural Indicators for Australia—First edition consultation draft*; National Centre for Culture and Recreation Statistics Australian Bureau of Statistics: Adelaide, Australia, 2010.
- 117. UNESCO. Towards a UNESCO Suite of Indicators on Literature Review; UNESCO: Paris, France, 2010.
- 118. Marla Waltman Daschko Conceptual Framework for Culture Statistics 2011; Department of Canadian Heritage: Gatineau, QC, Canada, 2011.
- 119. Ministry of Education and Culture of Finland. *Effectiveness Indicators to Strengthen the Knowledge Base for Cultural Policy;* Ministry of Education and Culture: Helsinki, Sweden, 2011.
- 120. ESSnet-CULTURE European Statistical System Network on Culture—Final Report; Ministry of Culture Finances and Cultural Statistics Department (CoStaC): Luxembourg, 2012.
- 121. Montalto, V. Measuring the economic impact of CCIs policies. TafterJournal 2012, 49, 1-4.

- 122. UNDP Creative Economy Report. *Widening Local Development Pathways 2013 Special Edition;* United Nations/UNDP: New York, NY, USA; UNESCO: Paris, France, 2013.
- 123. Columbia Basin Rural Development Institute. *Cultural Indicators Literature Review. Developing the Columbia Basin Rural Development Institute's Cultural Research Pillar;* Columbia Basin Rural Development Institute: Castlegar, BC, Canada, 2013.
- 124. El Beyrouty, K.; Tessler, A. Economic impact of the UK heritage tourism economy; Oxford Economics: Oxford, UK, 2010.
- 125. James, P. Assessing Cultural Sustainability: Agenda 21 for Culture; Agenda 21 for culture—Committee on culture of United Cities and LocalGovernments (UCLG): Barcelona, Spain, 2014.
- 126. Carnwath, J.D.; Brown, A.S. Understanding the Value and Impacts of Cultural Experiences; Arts Council England: Manchester, UK, 2014.
- 127. UNESCO. Culture for Development Indicators Suite (CDIS); UNESCO: Paris, France, 2014.
- 128. United Nations. Indicators for monitoring the Millennium Development Goals. Definitions, Rationale, Concepts and Sources; UN: New York, NY, USA, 2003; ISBN 92-1-161467-8.
- 129. Global network "Future we want includes Culture". *Recognizing the Role of Culture To Strengthen the Un Post-2015;* UNESCO: Paris, France, 2015.
- 130. Kushner, R.; Cohen, R. National Arts Index 2016: An Annual Measure of the Vitality of Arts and Culture in the United States: 2002–2013; Americans for the Arts: Washington, DC, USA, 2016.
- 131. Council of Europe. Cultural participation and inclusive societies; Council of Europe: Strasbourg, France, 2016.
- 132. Howland, A.; Ruckstuhl, S.; Sheehan, I.S.; Samuelson, L.; Bowman, L. Baltimore's Sustainable Future: Localizing the UN Sustainable Development Goals, Strategies and Indicators; Sustainable Development Solutions Network: Baltimore, MD, USA, 2016.
- 133. European Commission. *The European Tourism Indicator System: ETIS Toolkit for Sustainable Destination Management;* European Commission: Bruxelles, Belgium, 2016.
- 134. UNESCO. Measuring culture's contribution in Agenda 2030 Sustainable Development Goals; UNESCO: Paris, France, 2017.
- 135. Ortega-Villa, L.M.; Ley-Garcia, J. Analysis of Cultural Indicators: A Comparison of Their Conceptual Basis and Dimensions. *Soc. Indic. Res.* **2018**, 137, 413–439. [CrossRef]
- Ren, W.; Han, F. Indicators for assessing the sustainability of built heritage attractions: An Anglo-Chinese study. Sustainability 2018, 10, 2504. [CrossRef]
- 137. ISTAT. Cultura socialità e tempo libero; ISTAT: Rome, Italy, 2018.
- 138. Eurostat. Culture Statistics: 2019 Edition. Statistical Books; Eurostat: Bruxelles, Belgium, 2019; ISBN 978-92-76-09702-0. [CrossRef]
- 139. Eurostat. *Smarter, greener, more inclusive? INDICATORS TO SUPPORT THE EUROPE 2020 STRATEGY;* Publications Office of the European Union: Luxembourg, 2019; ISBN 978-92-76-09825-6.
- 140. UNESCO. Thematic Indicators for Culture in the 2030 Agenda; UNESCO: Paris, France, 2019.
- 141. Montalto, V.; Tacao Moura, C.J.; Langedijk, S.; Saisana, M. Culture counts: An empirical approach to measure the cultural and creative vitality of European cities. *Cities* **2019**, *89*, 167–185. [CrossRef]
- 142. Costanzo, L.; Ferrara, A.; Truglia, F.G. *BES: Paesaggio e patrimonio culturale. In Rapporto BES 2019: Il Benessere Equo e Sostenibile in Italia*; ISTAT: Rome, Italy, 2018; pp. 129–145.
- 143. European Commission Circular Economy Indicators. Available online: https://ec.europa.eu/environment/ecoap/indicators/ circular-economy-indicators_en (accessed on 22 April 2021).
- 144. Euroepan Commission. The Cultural and Creative Cities Monitor; Euroepan Commission: Brussels, Belgium, 2019.
- 145. ICOM; OECD. Culture and Local Development: Maximizing the Impact; ICOM: Paris, France; OECD: Paris, France, 2019.
- 146. Historic England. Heritage Counts. There's no place like old homes. Re-use and Recycle to Reduce Carbon; Historic England: London, UK, 2019.
- 147. Heisel, F.; Rau-Oberhuber, S. Calculation and evaluation of circularity indicators for the built environment using the case studies of UMAR and Madaster. J. Clean. Prod. 2020, 243, 118482. [CrossRef]